Author's response to reviews

Title: Improving Risk Factors for Diabetes Complications in Primary Care: a Protocol for an Organizational Intervention Grounded in Complexity Science

Authors:

Michael L. Parchman MD (parchman@uthscsa.edu)
Jacqueline A. Pugh MD (pugh@uthscsa.edu)
Steven D. Culler Ph.D. (sculler@sph.emory.edu)
Polly H. Noel Ph.D. (noelp@uthscsa.edu)
Nedal H. Arar Ph.D. (ararn@uthscsa.edu)
Raquel L. Romero MD (romeror0@uthscsa.edu)
Raymond F. Palmer Ph.D. (palmerr@uthscsa.edu)

Version: 4 Date: 17 January 2008

Author's response to reviews:

Here is the copy edited revision of the manuscript. thanks!